Author’s response to reviews

Title: Experience of treating high risk prostate hyperplasia patients with a HPS120 Laser

Authors:

Li-Jun Chen (lijunchencn@163.com)
Haixing Mai (haixingmai@163.com)
Li Zhao (lizhaoab@163.com)
Nan Qu (nanqua@yeah.net)
Yalin Wang (yalinwanga@163.com)
Chen Huang (chenhuangd@126.com)
Xuechao Li (xuechaolia@163.com)
Jinkai Dong (jinkaidongq@163.com)
Fei Tang (feitanga@126.com)
Biao Chen (biaochena@126.com)

Version: 2
Date: 30 October 2013

Author’s response to reviews: see over
Dear Editors:

On behalf of my co-authors, I am submitting the enclosed material “Experience of treating high risk prostate hyperplasia patients with a HPS120 Laser” for possible publication in BMC UROLOGY.

At present we also confront the following problems: There are more and more patients who need to undergo prostate surgery, but these kinds of patients always suffer from a wide variety of medical complications, so the risks when carrying out minimally invasive surgery are excessive, and under most circumstances it is only possible to carry out bladder fistulization or administer an oral drug to alleviate urinary retention or to improve the urination symptoms. Therefore, we carried out the 120 watt PVP surgery for the above-described special patients, and the longest follow up time was 24 months. The aim of this study was to investigate the effects and safety of 120 watt PVP surgery for the high risk prostate hyperplasia patients.

We certify that we have participated sufficiently in the work to take public responsibility for the appropriateness of the experimental design and method, and the collection, analysis, and interpretation of the data.

We have reviewed the final version of the manuscript and approve it for publication.

To the best of our knowledge and belief, this manuscript has not been published in whole or in part nor is it being considered for publication elsewhere.

Best Regards.
Yours Sincerely,

Lijun Chen,

Department of Urology,

307 Hospital of PLA,

Beijing, 100071,

China

Tel: +86 1066947321

Fax: +86 1083638301

Email: lijunchencn@163.com